\documentclass{article}

\setlength{\parindent}{0pt}

\begin{document}

\thispagestyle{empty}

\centerline{\textbf{Math 165: Honors Calculus I \hfill Fall 1999}}
\centerline{\textbf{Course Information}}

\bigskip
\textsc{Instructor:} Dennis M. Snow, 313 CCMB, 631-7597, snow.1@nd.edu

\bigskip
\textsc{Classrooms:} MWF 11:45 DeBartolo 119; H 11:00 DeBartolo 126

\bigskip
\textsc{Office Hours:} 2--3 MWF, or by appointment

\bigskip
\textsc{Text:} T.~Apostol, Calculus, Vol.\ I, 2nd Ed., Chapters 1--4

\bigskip
\textsc{Exams:} There will be two one-hour exams and a two-hour final
on the following dates:
\begin{center} \begin{tabular}{ll}
Exam I &Thursday, September 30\\
Exam II &Thursday, November 11\\
Final &Wednesday,  December 15, 8:00--10:00 A.M.\\
\end{tabular} \end{center}
Make-up exams will only be given for officially excused absences.

\bigskip
\textsc{Quizzes:} There will be a short quiz given on Thursday of
every week, except for exam weeks and the first/Thanksgiving week of
classes. Each quiz is worth 5 points.

\bigskip
\textsc{Assignments:} There will be regular assignments due on the class
day after they are assigned (Wednesday's assignment is due on Friday).
Points will be deducted for late assignments and may not be accepted
at all.

\bigskip
\textsc{Attendance:} You are expected to attend every class. Excessive
absences may result in lowering your grade and even failing the course.

\bigskip
\textsc{Honor Code:} The Honor Code is in effect for all exams, quizzes
and assignments. You are encouraged to work together on the assignments,
but copying in any form or submitting work done by others as your own
is a violation of the Honor Code.  A group of up to three students may
submit an assignment together, as long as the group members are clearly
identified and they all contribute to all parts of the assignment.

\bigskip
\textsc{Email \& Web Pages:}
Information about the course can be found on the course web pages,
webct.nd.edu:8900. Your username is the same as your afsid and your
password is your student ID number (you can change this later).  Regular
grade updates and other information will be sent to you by email. I
encourage you to see me in my office or contact me through email if you
have any questions or would like to discuss any aspect of the course.

\bigskip
\textsc{Grades:}  Grades will be based on a total of 500 points broken
down as follows:
\begin{center} \begin{tabular}{lcl}
Exam I\&II  &2 @ 100  &= 200\\
Final       &         &= 150\\
Quizzes     &10 @ 5   &=  50\\
Assignments &(scaled) &= 100\\
\end{tabular} \end{center}

\end{document}
